Author’s response to reviews

Title: One size does not fit all: HIV prevalence and correlates of risk for men who have sex with men, transgender women in multiple cities in Papua New Guinea

Authors:

Avi Hakim (hxv8@cdc.gov)
Kelsey Coy (nur1@cdc.gov)
Steven Badman (sbadman@kirby.unsw.edu.au)
Barne Willie (barne.willie@pngimr.org.pg)
Rebecca Narokobi (r_narokobi@hotmail.com)
Josephine Gabuzzi (josephine.gabuzzi@pngimr.org.pg)
Simon Pekon (simonpekon@gmail.com)
Martha Kupul (martha.kupul@pngimr.org.pg)
Parker Hou (parker.hou@pngimr.org.pg)
Herrick Aeno (herick.aeno9@gmail.com)
Ruthy Neo Boli (ruthy.neo@pngimr.org.pg)
Joshua Nembari (joshua.nembari@pngimr.org.pg)
Sophie Ase (sophie.ase@pngimr.org.pg)
Angelyne Amos-Kuma (angeline.amos@gmail.com)
Nick Dala (mnickdala@gmail.com)
Damian Weikum (ndi9@cdc.gov)
Steven Callens (Steven.Callens@uzgent.be)
John Kaldor (jkaldor@kirby.unsw.edu.au)
Andrew Vallely (avalley@kirby.unsw.edu.au)
Angela Kelly-Hanku (a.kelly@unsw.edu.au)
Author’s response to reviews:

We have added the following text in the Declarations section at the request of the editors:

"Written permission to participate was not obtained from the parents/guardians of the minors included in this study because the study’s populations are stigmatized for their behaviors and obtaining parent/guardian permission to participate would put the individuals at risk of violence, stigmatization, or harassment from their parent/guardian. The above referenced ethics committees approved this consent procedure for minors."